Author’s response to reviews

Title: Physical activity and motor skills in children attending 43 preschools: A cross-sectional study.

Authors:

   Line Grønholt Olesen (lgolesen@health.sdu.dk)
   Peter Lund Kristensen (PLKristensen@health.sdu.dk)
   Mathias Ried-Larsen (mried-larsen@health.sdu.dk)
   Anders Grøntved (agroentved@health.sdu.dk)
   Karsten Froberg (kfroberg@health.sdu.dk)

Version: 6  Date: 1 September 2014

Author’s response to reviews:

It is the Coverletter 01.09.2014.docx, that counts.